Author's response to reviews

Title: Syphilis screening in the antenatal care: a cross-sectional study from Botswana

Authors:

Maria Romoren (maria.romoren@medisin.uio.no)
Mafizur Rahman (mafiz@info.bw)

Version: 5 Date: 28 June 2006

Author's response to reviews: see over
Dear Editor,

MS: 1030473365985120

Syphilis screening in the antenatal care: a cross-sectional study from Botswana

We are happy to learn that *BMC International Health and Human Rights* will publish our revised manuscript without further revision. We have made the formatting changes as requested:

1. Affiliations: The city (Gaborone) was added to Dr Rahman’s affiliation. Otherwise we believe the format of the affiliations was provided as required.
2. Section headings: All section headings were changed from capitals to sentence case.

We would like to express that it has been a pleasure to submit this manuscript to the BioMed Central. The editorial management of the paper has been remarkably efficient, and the input from the reviewers was indeed valuable.

Yours sincerely

Maria Romoren    Mafizur Rahman